Author’s response to reviews

Title: The Cytotoxic Effect and Glucose Uptake Modulation of Baeckea Frutescens on Breast Cancer Cells

Authors:

Shazwin Hani Shahruzaman (shazwinhani91@yahoo.com)
Mohd Fazirul Mustafa (mfazirul@gmail.com)
Salfarina Ramli (salfarina2892@puncakalam.uitm.edu.my)
Subashani Maniam (subashni@yahoo.com)
Sharida Fakurazi (sharida@upm.edu.my)
Sandra Maniam (sandra@upm.edu.my)

Version: 2 Date: 31 Jul 2019

Author’s response to reviews:

Comment: The Background section was edited and we have reworded and paraphrased the Method section to reduce the overlap.